\begin{table}%T6
\par
\caption{\label{table:referV}List of references to Table~\ref{table:Vrot}.}
%\centering
\par
\begin{tabular}{ll|ll} 
\hline \hline
Abbr.        & Reference                                         & Abbr         & Reference                                         \\
 \hline
ABB          &  van Albada  et~al. (\cite{Albada1985})        & KN           &  Kuno   \& Nakai (\cite{Kuno1997})                             \\
ADS          &  Appleton et~al.  (\cite{Appleton1981})              & KS           &  Krumm  \& Salpeter (\cite{Krumm1979})                           \\
AMB          &  Amram  et~al. (\cite{Amram1994})          & KY           &  Kenney \& Young (\cite{Kenney1988})                              \\
AS           &  van Albada  \& Shane (\cite{Albada1975})                        & MM           &  Milliard  \& Marcelin (\cite{Milliard1981})                       \\
B            &  Bosma (\cite{Bosma1981})                                       & NN           &  Nishiyama  \& Nakai (\cite{Nishiyama2001})                         \\
Bu           &  Buta (\cite{Buta1988})                                        & NNK          &  Nishiyama et~al. (\cite{Nishiyama2001e})                    \\
BB           &  de Blok  \& Bosma (\cite{Blok2002})                           & OH           &  Ondrechen  \& van der Hulst (\cite{Ondrechen1989})                  \\
BBS          &  Begeman et~al.  (\cite{Begeman1991})                & P            &  Peterson (\cite{Peterson1978})                                     \\
BCH          &  Bosma  et~al. (\cite{Bosma1980})             & PCB          &  Puche et~al.  (\cite{Puche1990})                   \\
BHS          &  Bosma et~al. (\cite{Bosma1977})           & PCG          &  Puche et~al. (\cite{Puche1991})               \\
BM           &  Bajaja   \& Martin  (\cite{Bajaja1985})                         & R            &  Rots (\cite{Rots1980})                                        \\
BMR          &  Becker  et~al. (\cite{Becker1988})                & RCC          &  Rogstad et~al. (\cite{Rogstad1979})                    \\
BR           &  Broeils  \& Rhee (\cite{Broeils1997})                            & Re           &  Reakes (\cite{Reakes1979})                                       \\
BSC          &  Bravo-Alfaro  et~al. (\cite{Bravo1997})      & Ro           &  Rots (\cite{Rots1975})                                         \\
BWR          &  Bajaja  et~al. (\cite{Bajaja1995})         & RFT          &  Rubin et~al. (\cite{Rubin1980})                     \\
CCB          &  Carignan et~al. (\cite{Carignan1990})    & RKS          &  Ryder et~al. (\cite{Ryder2001})    \\
CD           &  Cox  \& Downes (\cite{Cox1996})                              & RMG          &  Reif  et~al. (\cite{Reif1982})                   \\
CG           &  Casertano  \& van Gorkom (\cite{Casertano1991})                    & S            &  Sage (\cite{Sage1993})                                         \\
CS           &  Corbelli \& Salucci (\cite{Corbelli2000})                         & Sc           &  Schneider (\cite{Schneider1989})                                    \\
DD           &  Dean  \& Davies (\cite{Dean1975})                             & SCR          &  Sperandio  et~al. (\cite{Sperandio1995})     \\
GDH          &  Garc\'{\i}a-Barreto et~al. (\cite{Garc1994})  & Sh           &  Shostak (\cite{Shostak1978})                                      \\
GGK          &  Guhathakurta  et~al. (\cite{Guhathakurta1988})    & SD           &  Staveley-Smith    \& Davies (\cite{Staveley1988})                  \\
GW           &  Gu\'{e}lin  \& Weliachew (1970)                    & SEC          &  Stark  et~al. (\cite{Stark1987})                  \\
H            &  Huchtmeier (\cite{Huchtmeier1975})                                  & SJL          &  Sandqvist et~al. (\cite{Sandqvist1995})     \\
HAS          &  Hoekstra et~al. (\cite{Hoekstra2001})            & SKB          &  Stark  et~al. (\cite{Stark1986})                   \\
HB           &  Helfer  \& Blitz (\cite{Helfer1995})                            & SNK          &  Sorai et~al. (\cite{Sorai2000})                    \\
HBS          &  Heckman et~al. (\cite{Heckman1978})                & TM           &  Thornley \& Mundy (\cite{Thornley1997})                            \\
HHG          &  Hewitt et~al. (\cite{Hewitt1983})              & vKS          &  van der Kruit  \& Shostak (\cite{Kruit1984})                    \\
HHM          &  Haynes  et~al. (\cite{Haynes1998})             & VPD          &  de Vaucouleurs et~al.  (\cite{Vaucouleurs1983})           \\
HR           &  Huchtmeier  \& Richter (\cite{Huchtmeier1986a})                     & Wa           &  Warmels (\cite{Warmels1988a})                                     \\
HRb          &  Huchtmeier  \& Richter (\cite{Huchtmeier1986b})                     & Wb           &  Warmels (\cite{Warmels1988b})                                     \\
HRG          &  Hunter et~al. (\cite{Hunter1986})                 & WAD          &  Wevers  et~al. (\cite{Wevers1984})             \\
HS           &  Huchtmeier  \& Seiradakis (1985)                   & WG           &  Weliachew   \& Gottesman (\cite{Weliachew1973})                     \\
HTR          &  Helfer   et~al. (\cite{Helfer2003})             & WKA          &  Wevers et~al. (\cite{Wevers1986})              \\
JM           &  J\"{o}rs\"{a}ter  \& van Moorsel (\cite{ter1995})            & YXT          &  Young  (\cite{Young1995})                       \\
KDI          &  Koper  et~al. (1991)                 & ZB           &  van Zee     \& Bryant (\cite{Zee1999}                       \\
      \hline
\end{tabular}
\end{table}